Author's response to reviews

Title: Type I Gaucher disease with exophthalmos and pulmonary arteriovenous malformation

Authors:

Chun-An Chen (chenca@ntu.edu.tw)
Nelson LS Tang (nelsontang@cuhk.edu.hk)
Yin-Hsiu Chien (chienyh@ha.mc.ntu.edu.tw)
Wei-Min Zhang (nelsontang@cuhk.edu.hk)
Jou-Kou Wang (jkw@ha.mc.ntu.edu.tw)
Wuh-Liang Hwu (hwu@ha.mc.ntu.edu.tw)

Version: 5 Date: 23 May 2005

Author's response to reviews:

1. Title page - The corresponding author was indicated. The heading 'case report' has been removed from the title.
2. Affiliations - Each affiliation has been written in full name according to the suggested format.
3. Keywords section has been removed.
4. The section of "Authors' contributions" has been revised according to the request.
5. References - All references have been changed to fit the format.
6. Patient's consent - English version of patient's consent has been uploaded.